Author's response to reviews

Title: Intravenous Vitamin C Administration Reduces Fatigue in Office Workers: A Double-blind Randomized Controlled Trial

Authors:

Sang-Yeon Suh (lisasuhmd@hotmail.com)
Woo Kyung Bae (naslig@hanmail.net)
Hong-Yup Ahn (ahn@dongguk.edu)
Sung-Eun Choi (c6300@hanmail.net)
Gyou-Chul Jung (chyjung2@hanmail.net)
Chang Hwan Yeom (lymphych@hanmail.net)

Version: 3 Date: 17 January 2012

Author's response to reviews: see over
Dear Editorial Team,

We appreciate your effort to improve our article.

We made changes as follows:

1. Affiliations: We gave number 1 and 2 to each affiliation you mentioned. And we updated the rest of the affiliations to be in numerical order. All affiliations are connected to the authors.

2. Email addresses for all the authors are listed on the title page using the suggested format.

3. All word/figure/table counts were removed throughout the manuscript.

4. We removed the 'Running/short title' from the title page.

5. We removed all tracking or highlighting from the manuscript.

6. Abbreviations were formatted to a paragraph.

7. The funding statement was moved from the conflicts of interests section. We provided this information in the acknowledgements section.

8. We listed the source of funding for the study in the acknowledgements section. The funding was not given for each author. The funding had no role for the manuscript preparation.

9. Conflicts of interests was placed after the reference list.

* We could not remove the title from the figures. We’ll be grateful if you make change for us.